Author’s response to reviews

Title: Leadership, institution building and pay-back of health systems research in Mexico

Authors:

Miguel Angel Gonzalez-Block (mgonzalezblock@insp.mx)

Version: 6 Date: 25 September 2009

Author's response to reviews:

"FUNSALUD" was changed for "Funsalud" to make it coherent throughout the text.